Author's response to reviews

Title: Semi-Purified Extracts of Commelina benghalensis (Commelinaceae) Induce Apoptosis and Cell Cycle Arrest in Jurkat-T Cells

Authors:

Kgomotso W Lebogo (lebogokw@webmail.co.za)
Matlou P Mokgotho (matlou.mokgotho@ul.ac.za)
Victor P Bagla (drvictorb@yahoo.com)
Thabe M Matsebatlela (thabe.matsebatlela@ul.ac.za)
Vusi G Mbazima (vusi.mbazima@ul.ac.za)
Leshwene J Shai (shailj@tut.ac.za)
Leseilame J Mampuru (leseilane.mampuru@ul.ac.za)

Version: 3
Date: 10 December 2013

Author's response to reviews: see over
To: The Editor

Subject: Response to Reviewer’s Comments – MS: 1705341711105331

Dear Editor

The authors of MS: 1705341711105331 have looked at the article and addressed the suggested comments.

• **Reviewer 1:**
  
  No adjustments requested.

• **Reviewer 2:**

  Under Abstract:
  
  Amendments were implemented as requested.

  Under Introduction:
  
  Amendments were implemented as requested.

  Under Methods:
  
  Amendments were implemented as requested.

  Under Results:
  
  Amendments were implemented as requested.
Under Conclusion:

Amendments were implemented as requested.

The authors hope that the amendments will enhance the quality of the manuscript.

Thanking you in advance

Kind regards

Prof MP Mokgotho

Associate Professor and Researcher